\begin{table}%t1
%\centering
\par
\caption{\label{table1}Observed and inferred stellar parameters for the 
stars hosting planets presented in this paper. Definitions and
sources of the quoted values are given in the text. The age and
rotational period estimates are based on calibrations of the 
$R^\prime_{\rm HK}$ activity indicator \citep{Donahue-93,Noyes-84}, 
whose reference source is also indicated: (S) for this paper following 
\citet{Santos-2000:b}, (H) for \citet{Henry-96} and~(B) for 
\citet{Butler-2002}.  The applied analyses and uncertainty estimates 
can be found in the quoted references.}
\begin{tabular}{lllccccccrccllr}
\hline\hline
HD       &Sp     &$V$ &$B-V$  &$\pi$  &$M_V$ &$L$ 
&$T_{\rm eff}$   &$\log{g}$   &\multicolumn{1}{l}{\hspace*{-1mm}[Fe/H]}  
&$M_\star$  &$v\sin i$  
&\hspace*{-2mm}$\log(R^{\prime}_{\rm HK})$ &age%($R^{\prime}_{HK}$) 
&\multicolumn{1}{l}{\hspace*{-2mm}$P_{\rm rot}$}\\ %(R^{\prime}_{HK})$ \\
 & & & &[mas] & &[$L_{\odot}]$ &[K] &[cgs] & &[$M_{\odot}$]
 &[km~s$^{-1}$] & &[Gy] &\hspace*{-2mm}[day] \\
\hline
6434         &G3IV  &\hspace*{-1mm}7.72 &0.613 &24.80 &4.69 &1.12
    &5835    &4.60  &\hspace*{-1mm}$-0.52$ &0.79   &2.3
    &\hspace*{-2mm}$-4.89$ (H) &3.8   &\hspace*{-2mm}18.6 \\           
%10647        &F8V   &5.52    &0.551  &57.63  &4.32   &1.51
%    &6143    &4.48  &$-0.03$ &1.14   &4.88
%    &$-4.69$ &1.8   &8.8  \\
19994        &F8V   &\hspace*{-1mm}5.07 &0.575  &44.69  &3.32   &3.81
    &6217    &4.29  &\hspace*{-1mm}0.25    &1.34   &8.1
    &\hspace*{-2mm}$-4.77$ (S) &2.4   &\hspace*{-2mm}12.2 \\
65216        &G5V   &\hspace*{-1mm}7.97 &0.672  &28.10  &5.21   &0.71
    &5666    &4.53  &\hspace*{-1mm}$-0.12$ &0.92   &$<$1
    &\multicolumn{1}{c}{\hspace*{-2mm}--}  &\multicolumn{1}{c}{--} 
    &\multicolumn{1}{c}{\hspace*{-2mm}--}  \\
82943        &G0    &\hspace*{-1mm}6.54 &0.623  &36.42  &4.35   &1.50
    &6005    &4.45  &\hspace*{-1mm}0.32 &1.15   &1.7
    &\hspace*{-2mm}$-4.82$ (S) &2.9   &\hspace*{-2mm}18.0   \\
83443        &K0V   &\hspace*{-1mm}8.23 &0.811  &22.97  &5.04   &0.88
    &5454    &4.33  &\hspace*{-1mm}0.35 &0.90   &1.4
    &\hspace*{-2mm}$-4.85$ (B) &3.2   &\hspace*{-2mm}35.3 \\
92788        &G5    &\hspace*{-1mm}7.31 &0.694  &30.94  &4.76   &1.05
    &5821    &4.45  &\hspace*{-1mm}0.32 &1.10   &1.8
    &\hspace*{-2mm}$-4.73$ (S) &2.1   &\hspace*{-2mm}21.3 \\
111232       &G5V   &\hspace*{-1mm}7.59 &0.701  &34.63  &5.29   &0.69
    &5494    &4.50  &\hspace*{-1mm}$-0.36$ &0.78   &1.2
    &\hspace*{-2mm}$-4.98$ (H) &5.2  &\hspace*{-2mm}30.7 \\ 
114386       &K3V   &\hspace*{-1mm}8.73 &0.982  &35.66  &6.49   &0.29
    &4804    &4.36  &\hspace*{-1mm}$-0.08$    &0.68   &1.0
    &\multicolumn{1}{c}{\hspace*{-2mm}--}  &\multicolumn{1}{c}{--} 
    &\multicolumn{1}{c}{\hspace*{-2mm}--}\\ 
121504       &G2V   &\hspace*{-1mm}7.54 &0.593  &22.54  &4.30   &1.55
    &6075    &4.64  &\hspace*{-1mm}0.16 &1.18   &2.6
    &\hspace*{-2mm}$-4.57$ (S) &1.2   &\hspace*{-2mm}8.6  \\ 
142415       &G1V   &\hspace*{-1mm}7.33 &0.621  &28.93  &4.64   &1.14
    &6045    &4.53  &\hspace*{-1mm}0.21 &1.03   &3.3
    &\hspace*{-2mm}$-4.55$ (S) &1.1  &\hspace*{-2mm}9.6  \\ 
147513       &G3/5V &\hspace*{-1mm}5.37 &0.625  &77.69  &4.82   &0.98
    &5883    &4.51  &\hspace*{-1mm}0.06 &1.11   &1.5
    &\hspace*{-2mm}$-4.38$ (S) &0.3   &\hspace*{-2mm}4.7  \\ 
169830       &F8V   &\hspace*{-1mm}5.90 &0.517  &27.53  &3.10   &4.59
    &6299    &4.10  &\hspace*{-1mm}0.21    &1.40   &3.3
    &\hspace*{-2mm}$-4.82$ (S) &2.8   &\hspace*{-2mm}8.3 \\ 
196050       &G3V   &\hspace*{-1mm}7.50 &0.667  &21.31  &4.14   &1.83
    &5918    &4.34  &\hspace*{-1mm}0.22    &1.10   &3.1 
    &\hspace*{-2mm}$-4.65$ (S) &1.6   &\hspace*{-2mm}16.0 \\
216437      &G4IV/V &\hspace*{-1mm}6.04 &0.660  &37.71  &3.92   &2.25
    &5887    &4.30  &\hspace*{-1mm}0.25    &1.06   &2.5
    &\hspace*{-2mm}$-5.01$ (H) &5.8   &\hspace*{-2mm}26.7 \\
216770       &K0V   &\hspace*{-1mm}8.11 &0.821  &26.39  &5.22   &0.79
    &--      &--    &\hspace*{-1mm}0.23    &0.90  &1.4
    &\hspace*{-2mm}$-4.84$ (H) &3.1   &\hspace*{-2mm}35.6 \\ 
\hline
\end{tabular}
\end{table}